Author's response to reviews

Title: Health profiles of 996 melanoma survivors: the M. D. Anderson experience

Authors:

Charles Stava (cstava@mdanderson.org)
Martha Beck (mbeck@mdanderson.org)
L. Todd Weiss (ltweiss@mdanderson.org)
Adriana Lopez (adrilopez@mdanderson.org)
Rena Vassilopoulou-Sellin (rsellin@mdanderson.org)

Version: 7 Date: 12 April 2006

Author's response to reviews:

April 12, 2006

Peter Newmark, MD, Editor-in-chief

Dear Dr. Newmark,

Thank you for the opportunity to publish our article in your esteemed journal. Please find the following changes to the manuscript and submission, as per your request.

Title & Abstract - Ensured these are identical in the manuscript and in the submission system.
Author list - The list of authors in the manuscript are written exactly as they are in the submission system.
Qualifications - We have removed the authors' qualifications from the title page.
Running title - Has been removed from the title page.
Section headings - Used sentence case for all headings and sub-headings in the manuscript.
Table Titles - Changed to sentence case.
Figure title - The titles and numbers have been removed from the images.
Website addresses - Website addresses have been removed from the manuscript text.

Sincerely yours,

Rena Vassilopoulou-Sellin, M.D.
Professor of Internal Medicine and Pediatrics
Department of Endocrine Neoplasia and Hormonal Disorders
The University of Texas M. D. Anderson Cancer Center